Author’s response to reviews

Title: Recruitment challenges in clinical research. A case study and lessons learned.

Authors:

Karin Sygna (karin.sygna@rr-research.no)
Safora Johansen (sjh@ous-hf.no)
Cornelia Ruland (cornelia.ruland@rr-research.no)

Version: 2 Date: 18 March 2015

Author’s response to reviews:

1) included trial registration number, along with the date of registration, at the end of the Abstract.

2) included a statement of ethical approval in the Methods section of the manuscript, including the name of the body which gave approval.

3) no need for list of abbreviations

4) Each author is individually mentioned in the Authors’ Contributions section.

5) Acknowledgements section is included at the end of the manuscript before the reference list.